Author’s response to reviews

Title: Moyamoya Disease: A Clinical Spectrum Literature Review and Case series from a Tertiary care hospital in Pakistan.

Authors:

Sana Shoukat (sana.shoukat@gmail.com)
Ahmed Itrat (ahmeditrat@gmail.com)
Moazzam Zaidi (drmoazzam_2003@yahoo.com)
Ather Taqui (ather.taqui@gmail.com)
Ayeesha Kamal (ayeeshakamal@aku.edu)

Version: 4 Date: 8 April 2009

Author’s response to reviews: see over
March 7, 2009

**MS: 1732362869224630**

Dear Editor,

As per request the following have been made

Major revisions (We require the author to make these changes),
   Have done so in the first round.

Figure legends - please add a `Figure legends¿ section for the figure title and legend information. Place this section in a position after the references. Please remove the title and legend text from the figure images themselves.
   Done

Figure citations - please cite each figure within the text in numerical order.
   Done

*All the below have been done except access to the author list which I don’t seem to get.*

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study. To `The title of my prestigious publication: a scientific study'

Country - Please remember to include the country in the affiliation details.

Email address - please include email addresses for all the authors on the title page. Please use the following format-
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk

Disclosure of interest - please move this information to the 'Competing interests' section just before the references.
Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page. Please remove the keywords from this section also.

Patients and methods - please rename this section "Methods".

Tables - please move to a position after the references.

Table titles - please move these to above the corresponding table.

Thank you

Ayeesha Kamal, MD, FAHA
Assistant Professor Neurology
AKUH, Karachi
Pakistan